**Author's response to reviews**

**Title:** Interventions aimed at improving the nursing work environment: a systematic review

**Authors:**

Donna MJ Schalk (donna.schalk@han.nl)
Marloes LP Bijl (m.bijl@student.maastrichtuniversity.nl)
Greta G Cummings (gretac@ualberta.ca)
Ruud JG Halfens (r.halfens@zw.unimaas.nl)
Louk Hollands (l.hollands@zw.unimaas.nl)

**Version:** 9  **Date:** 21 April 2010

**Author's response to reviews:**

Thank you for copyediting this manuscript and providing us with details edits that we needed to make. We have completed all requested changes which includes the following:

**Major revisions**

Tables: We have moved two tables to Additional files, and corrected all in text citations accordingly. We have created a legend for additional files as requested.

**Minor revisions**

References: We have added all authors to the reference list up to 30 authors names for each citation, and removed et al.

Figures: We have removed the title from Figure 1 and added a Figure legend as requested.

Typography: We have reviewed the manuscript one more time for typo corrections.

Please advise if there are any further changes that we need to make.

thank you

Greta Cummings